\documentclass{ws-procs9x6}

\usepackage{booktabs}

\begin{document}
\begin{thebibliography}{10}
\providecommand*{\bibinfo}[2]{#2}
\providecommand*{\eprint}[1]{#1}
\providecommand*{\url}[1]{#1}
\bibitem{landauer}
\bibinfo{author}{R.~Landauer}, \bibinfo{journal}{Physics Today}
  \bibinfo{volume}{\textbf{44}}(5), \bibinfo{pages}{23} (\bibinfo{date}{May
  1991}), \url{https://doi.org/10.1063/1.881299}.
\bibitem{copeland04turing}
\bibinfo{author}{A.~M. Turing}, in \bibinfo{editors}{B.~J. Copeland}, ed.,
  \emph{The Essential {T}uring} (\bibinfo{publisher}{Oxford University Press},
  Oxford and New York, \bibinfo{year}{2004}),
  \url{https://global.oup.com/academic/product/the-essential-turing-9780198250807}.
\bibitem{Arthaud-en}
\bibinfo{author}{A.~Artaud}, \bibinfo{title}{\emph{The Theatre and Its Double}}
  (\bibinfo{publisher}{Alma Classics Limited}, Richmond Surrey, UK,
  \bibinfo{year}{2010 (1938, 1964)}), translated by Victor Corti,
  \url{https://almabooks.com/product/the-theatre-and-its-double/}.
\bibitem{bernais-propaganda}
\bibinfo{author}{E.~Bernays}, \bibinfo{title}{\emph{Propaganda}}
  (\bibinfo{publisher}{Routledge}, \bibinfo{year}{1928}),
  \url{http://self.gutenberg.org/Get956uFile.aspx?&bookid=100002651}.
\bibitem{clauser-talkvie}
\bibinfo{author}{J.~F. Clauser}, in \bibinfo{editors}{R.~Bertlmann and
  A.~Zeilinger}, eds., \emph{Quantum (Un)speakables: {F}rom {B}ell to Quantum
  Information} (\bibinfo{publisher}{Springer}, Berlin, \bibinfo{year}{2002}),
  \bibinfo{pages}{pp. 61--96}, ISBN
  \bibinfo{isbn}{978-3-662-05032-3,978-3-540-42756-8},
  \url{https://doi.org/10.1007/978-3-662-05032-3\_6}.
\bibitem{wigner}
\bibinfo{author}{E.~P. Wigner}, \bibinfo{journal}{Communications on Pure and
  Applied Mathematics} \bibinfo{volume}{\textbf{13}}, \bibinfo{pages}{1}
  (\bibinfo{date}{1960}), \url{https://doi.org/10.1002/cpa.3160130102}.
\bibitem{Exner-1908}
\bibinfo{author}{F.~S. Exner}, \bibinfo{title}{\emph{{\"U}ber {G}esetze in
  {N}aturwissenschaft und {H}umanistik: {I}naugurationsrede gehalten am 15.
  {O}ktober 1908}} (\bibinfo{publisher}{H\"older, Ebooks on Demand
  Universit\"atsbibliothek Wien}, Vienna, \bibinfo{year}{1909, 2016}), handle
  https://hdl.handle.net/11353/10.451413, o:451413, Uploaded: 30.08.2016,
  \url{http://phaidra.univie.ac.at/o:451413}.
\bibitem{Yanofsky-chaos}
\bibinfo{author}{N.~S. Yanofsky}, \bibinfo{title}{\emph{Finding order in
  chaos}} (\bibinfo{date}{2017}), preprint, received March 20th, 2017,
  \url{http://www.sci.brooklyn.cuny.edu/~noson/FindingOrder.pdf}.
\bibitem{svozil-2018-was}
\bibinfo{author}{C.~S. Calude} and \bibinfo{author}{K.~Svozil},
  \bibinfo{journal}{Philosophies} \bibinfo{volume}{\textbf{4}}(2),
  \bibinfo{pages}{17} (\bibinfo{date}{2019}), \eprint{arXiv:1812.04416},
  \url{https://doi.org/10.3390/philosophies4020017}.
\bibitem{zuse-70}
\bibinfo{author}{K.~Zuse}, \bibinfo{title}{\emph{Calculating Space. MIT
  Technical Translation AZT-70-164-GEMIT}} (\bibinfo{publisher}{MIT (Proj.
  MAC)}, Cambridge, MA, \bibinfo{year}{1970}).
\bibitem{toffoli:79}
\bibinfo{author}{T.~Toffoli}, in \bibinfo{editors}{G.~J. Klir}, ed.,
  \emph{Applied General Systems Research: Recent Developments and Trends}
  (\bibinfo{publisher}{Plenum Press, Springer US}, New York, London, and
  Boston, MA, \bibinfo{year}{1978}), \bibinfo{pages}{pp. 395--400}, ISBN
  \bibinfo{isbn}{978-1-4757-0555-3},
  \url{https://doi.org/10.1007/978-1-4757-0555-3\_29}.
\bibitem{simula}
\bibinfo{author}{D.~F. Galouye}, \bibinfo{title}{\emph{Simulacron 3}}
  (\bibinfo{publisher}{Bantam Books}, New York, \bibinfo{year}{1964}).
\bibitem{permutationcity}
\bibinfo{author}{G.~Egan}, \bibinfo{title}{\emph{Permutation City}}
  (\bibinfo{year}{1994}), ISBN \bibinfo{isbn}{006105481X,9780061054815},
  accessed on January 4, 2017,
  \url{http://www.gregegan.net/PERMUTATION/Permutation.html}.
\bibitem{Bostrom-sim}
\bibinfo{author}{N.~Bostrom}, \bibinfo{journal}{The Philosophical Quarterly}
  \bibinfo{volume}{\textbf{53}}(211), \bibinfo{pages}{243}
  (\bibinfo{date}{2003}),
  \eprint{https://www.simulation-argument.com/simulation.pdf},
  \url{https://doi.org/10.1111/1467-9213.00309}.
\bibitem{svozil-2016-quantum-hokus-pokus}
\bibinfo{author}{K.~Svozil}, \bibinfo{journal}{Ethics in Science and
  Environmental Politics (ESEP)} \bibinfo{volume}{\textbf{16}}(1),
  \bibinfo{pages}{25} (\bibinfo{date}{2016}), \eprint{arXiv:1605.08569},
  \url{https://doi.org/10.3354/esep00171}.
\bibitem{v-neumann-49}
\bibinfo{author}{J.~{von Neumann}}, \bibinfo{title}{\emph{{M}athematische
  {G}rundlagen der {Q}uantenmechanik}} (\bibinfo{publisher}{Springer}, Berlin,
  Heidelberg, \bibinfo{year}{1932, 1996}), 2nd ed., ISBN
  \bibinfo{isbn}{978-3-642-61409-5,978-3-540-59207-5,978-3-642-64828-1},
  {E}nglish translation in~\cite{v-neumann-55},
  \url{https://doi.org/10.1007/978-3-642-61409-5}.
\bibitem{schrodinger}
\bibinfo{author}{E.~Schr{\"{o}}dinger}, \bibinfo{journal}{Naturwissenschaften}
  \bibinfo{volume}{\textbf{23}}, \bibinfo{pages}{807} (\bibinfo{date}{1935}),
  \url{https://doi.org/10.1007/BF01491891, https://doi.org/10.1007/BF01491914,
  https://doi.org/10.1007/BF01491987}.
\bibitem{schroedinger-interpretation}
\bibinfo{author}{E.~Schr{\"{o}}dinger}, \bibinfo{title}{\emph{The
  Interpretation of Quantum Mechanics. {D}ublin Seminars (1949-1955) and Other
  Unpublished Essays}} (\bibinfo{publisher}{Ox Bow Press}, Woodbridge,
  Connecticut, \bibinfo{year}{1995}).
\bibitem{everett}
\bibinfo{author}{H.~{Everett III}}, \bibinfo{journal}{Reviews of Modern
  Physics} \bibinfo{volume}{\textbf{29}}, \bibinfo{pages}{454}
  (\bibinfo{date}{1957}), \url{https://doi.org/10.1103/RevModPhys.29.454}.
\bibitem{everett-1956}
\bibinfo{author}{H.~{Everett III}}, in \bibinfo{editors}{J.~A. Barrett and
  P.~Byrne}, eds., \emph{The {E}verett Interpretation of Quantum Mechanics:
  Collected Works 1955-1980 with Commentary} (\bibinfo{publisher}{Princeton
  University Press}, Princeton, NJ, \bibinfo{year}{1956,2012}),
  \bibinfo{pages}{pp. 72--172}, ISBN \bibinfo{isbn}{9780691145075},
  \url{http://press.princeton.edu/titles/9770.html}.
\bibitem{wigner:mb}
\bibinfo{author}{E.~P. Wigner}, in \bibinfo{editors}{I.~J. Good}, ed.,
  \emph{The Scientist Speculates} (\bibinfo{publisher}{Heinemann, Basic Books,
  and Springer-Verlag}, London, New York, and Berlin, \bibinfo{year}{1961,
  1962, 1995}), \bibinfo{pages}{pp. 284--302},
  \url{https://doi.org/10.1007/978-3-642-78374-6\_20}.
\bibitem{london-Bauer-1983}
\bibinfo{author}{F.~London} and \bibinfo{author}{E.~Bauer}, in \emph{Quantum
  Theory and Measurement} (\bibinfo{publisher}{Princeton University Press},
  Princeton, NJ, \bibinfo{year}{1983}), \bibinfo{pages}{pp. 217--259},
  consolidated translation of French original~\cite{london-Bauer-1939}.
\bibitem{mermin-07}
\bibinfo{author}{D.~N. Mermin}, \bibinfo{title}{\emph{Quantum Computer
  Science}} (\bibinfo{publisher}{Cambridge University Press}, Cambridge,
  \bibinfo{year}{2007}), ISBN \bibinfo{isbn}{9780521876582},
  \url{https://doi.org/10.1017/CBO9780511813870}.
\bibitem{PhysRevD.22.879}
\bibinfo{author}{A.~Peres}, \bibinfo{journal}{Physical Review D}
  \bibinfo{volume}{\textbf{22}}(4), \bibinfo{pages}{879} (\bibinfo{date}{Aug
  1980}), \url{https://doi.org/10.1103/PhysRevD.22.879}.
\bibitem{PhysRevA.25.2208}
\bibinfo{author}{M.~O. Scully} and \bibinfo{author}{K.~Dr\"uhl},
  \bibinfo{journal}{Physical Review A} \bibinfo{volume}{\textbf{25}}(4),
  \bibinfo{pages}{2208} (\bibinfo{date}{April 1982}),
  \url{https://doi.org/10.1103/PhysRevA.25.2208}.
\bibitem{greenberger2}
\bibinfo{author}{D.~M. Greenberger} and \bibinfo{author}{A.~YaSin},
  \bibinfo{journal}{Foundation of Physics} \bibinfo{volume}{\textbf{19}}(6),
  \bibinfo{pages}{679} (\bibinfo{date}{1989}),
  \url{https://doi.org/10.1007/BF00731905}.
\bibitem{Nature351}
\bibinfo{author}{M.~O. Scully}, \bibinfo{author}{B.-G. Englert}, and
  \bibinfo{author}{H.~Walther}, \bibinfo{journal}{Nature}
  \bibinfo{volume}{\textbf{351}}, \bibinfo{pages}{111} (\bibinfo{date}{May
  1991}), \url{https://doi.org/10.1038/351111a0}.
\bibitem{Zajonc-91}
\bibinfo{author}{A.~G. Zajonc}, \bibinfo{author}{L.~J. Wang},
  \bibinfo{author}{X.~Y. Zou}, and \bibinfo{author}{L.~Mandel},
  \bibinfo{journal}{Nature} \bibinfo{volume}{\textbf{353}},
  \bibinfo{pages}{507} (\bibinfo{date}{October 1991}),
  \url{https://doi.org/10.1038/353507b0}.
\bibitem{PhysRevA.45.7729}
\bibinfo{author}{P.~G. Kwiat}, \bibinfo{author}{A.~M. Steinberg}, and
  \bibinfo{author}{R.~Y. Chiao}, \bibinfo{journal}{Physical Review A}
  \bibinfo{volume}{\textbf{45}}(11), \bibinfo{pages}{7729} (\bibinfo{date}{Jun
  1992}), \url{https://doi.org/10.1103/PhysRevA.45.7729}.
\bibitem{PhysRevLett.73.1223}
\bibinfo{author}{T.~Pfau}, \bibinfo{author}{S.~Sp\"alter},
  \bibinfo{author}{C.~Kurtsiefer}, \bibinfo{author}{C.~R. Ekstrom}, and
  \bibinfo{author}{J.~Mlynek}, \bibinfo{journal}{Physical Review Letters}
  \bibinfo{volume}{\textbf{73}}(9), \bibinfo{pages}{1223} (\bibinfo{date}{Aug
  1994}), \url{https://doi.org/10.1103/PhysRevLett.73.1223}.
\bibitem{PhysRevLett.75.3783}
\bibinfo{author}{M.~S. Chapman}, \bibinfo{author}{T.~D. Hammond},
  \bibinfo{author}{A.~Lenef}, \bibinfo{author}{J.~Schmiedmayer},
  \bibinfo{author}{R.~A. Rubenstein}, \bibinfo{author}{E.~Smith}, and
  \bibinfo{author}{D.~E. Pritchard}, \bibinfo{journal}{Physical Review Letters}
  \bibinfo{volume}{\textbf{75}}(21), \bibinfo{pages}{3783}
  (\bibinfo{date}{November 1995}),
  \url{https://doi.org/10.1103/PhysRevLett.75.3783}.
\bibitem{hkwz}
\bibinfo{author}{T.~J. Herzog}, \bibinfo{author}{P.~G. Kwiat},
  \bibinfo{author}{H.~Weinfurter}, and \bibinfo{author}{A.~Zeilinger},
  \bibinfo{journal}{Physical Review Letters} \bibinfo{volume}{\textbf{75}}(17),
  \bibinfo{pages}{3034} (\bibinfo{date}{1995}),
  \url{https://doi.org/10.1103/PhysRevLett.75.3034}.
\bibitem{bell-a}
\bibinfo{author}{J.~S. Bell}, \bibinfo{journal}{Physics World}
  \bibinfo{volume}{\textbf{3}}, \bibinfo{pages}{33} (\bibinfo{date}{1990}),
  \url{https://doi.org/10.1088/2058-7058/3/8/26}.
\bibitem{engrt-sg-I}
\bibinfo{author}{B.-G. Englert}, \bibinfo{author}{J.~Schwinger}, and
  \bibinfo{author}{M.~O. Scully}, \bibinfo{journal}{Foundations of Physics}
  \bibinfo{volume}{\textbf{18}}(10), \bibinfo{pages}{1045}
  (\bibinfo{date}{1988}), \url{https://doi.org/10.1007/BF01909939}.
\bibitem{engrt-sg-II}
\bibinfo{author}{J.~Schwinger}, \bibinfo{author}{M.~O. Scully}, and
  \bibinfo{author}{B.-G. Englert}, \bibinfo{journal}{Zeitschrift f{\"u}r Physik
  D: Atoms, Molecules and Clusters} \bibinfo{volume}{\textbf{10}}(2-3),
  \bibinfo{pages}{135} (\bibinfo{date}{1988}),
  \url{https://doi.org/10.1007/BF01384847}.
\bibitem{Myrvold2011237}
\bibinfo{author}{W.~C. Myrvold}, \bibinfo{journal}{Studies in History and
  Philosophy of Science Part B: Studies in History and Philosophy of Modern
  Physics} \bibinfo{volume}{\textbf{42}}(4), \bibinfo{pages}{237}
  (\bibinfo{date}{2011}), \url{https://doi.org/10.1016/j.shpsb.2011.07.001}.
\bibitem{feynman-law}
\bibinfo{author}{R.~P. Feynman}, \bibinfo{title}{\emph{The Character of
  Physical Law}} (\bibinfo{publisher}{MIT Press}, Cambridge, MA,
  \bibinfo{year}{1965}), ISBN \bibinfo{isbn}{9780262060165,9780262560030},
  \url{https://mitpress.mit.edu/books/character-physical-law}.
\bibitem{mermin-1989-shutup}
\bibinfo{author}{D.~N. Mermin}, \bibinfo{journal}{Physics Today}
  \bibinfo{volume}{\textbf{42}}, \bibinfo{pages}{9} (\bibinfo{date}{1989}),
  \url{https://doi.org/10.1063/1.2810963}.
\bibitem{mermin-2004-shutup}
\bibinfo{author}{D.~N. Mermin}, \bibinfo{journal}{Physics Today}
  \bibinfo{volume}{\textbf{57}}, \bibinfo{pages}{10} (\bibinfo{date}{1989}),
  \url{https://doi.org/10.1063/1.1768652}.
\bibitem{zeil-05_nature_ofQuantum}
\bibinfo{author}{A.~Zeilinger}, \bibinfo{journal}{Nature}
  \bibinfo{volume}{\textbf{438}}, \bibinfo{pages}{743} (\bibinfo{date}{2005}),
  \url{https://doi.org/10.1038/438743a}.
\bibitem{nielsen-book10}
\bibinfo{author}{M.~A. Nielsen} and \bibinfo{author}{I.~L. Chuang},
  \bibinfo{title}{\emph{Quantum Computation and Quantum Information}}
  (\bibinfo{publisher}{Cambridge University Press}, Cambridge,
  \bibinfo{year}{2010}), 10th Anniversary Edition,
  \url{https://doi.org/10.1017/CBO9780511976667}.
\bibitem{svozil-2009-howto}
\bibinfo{author}{K.~Svozil}, \bibinfo{journal}{Physical Review A}
  \bibinfo{volume}{\textbf{79}}(5), \bibinfo{pages}{054306},
  \bibinfo{eid}{054306} (\bibinfo{numpages}{3}  pages)  (\bibinfo{date}{2009}),
  \eprint{arXiv:quant-ph/0903.2744},
  \url{https://doi.org/10.1103/PhysRevA.79.054306}.
\bibitem{svozil-2017-b}
\bibinfo{author}{K.~Svozil}, \bibinfo{journal}{Entropy}
  \bibinfo{volume}{\textbf{22}}(6), \bibinfo{pages}{602} (\bibinfo{date}{May
  2020}), \eprint{arXiv:1707.08915}, \url{https://doi.org/10.3390/e22060602}.
\bibitem{kochen1}
\bibinfo{author}{S.~Kochen} and \bibinfo{author}{E.~P. Specker},
  \bibinfo{journal}{Journal of Mathematics and Mechanics (now Indiana
  University Mathematics Journal)} \bibinfo{volume}{\textbf{17}}(1),
  \bibinfo{pages}{59} (\bibinfo{date}{1967}),
  \url{https://doi.org/10.1512/iumj.1968.17.17004}.
\bibitem{froissart-81}
\bibinfo{author}{M.~Froissart}, \bibinfo{journal}{Il Nuovo Cimento B
  (1971-1996)} \bibinfo{volume}{\textbf{64}}, \bibinfo{pages}{241}
  (\bibinfo{date}{1981}), 10.1007/BF02903286,
  \url{https://doi.org/10.1007/BF02903286}.
\bibitem{pitowsky:218}
\bibinfo{author}{I.~Pitowsky}, \bibinfo{journal}{Journal of Mathematical
  Physics} \bibinfo{volume}{\textbf{39}}(1), \bibinfo{pages}{218}
  (\bibinfo{date}{1998}), \url{https://doi.org/10.1063/1.532334}.
\bibitem{2015-AnalyticKS}
\bibinfo{author}{A.~A. Abbott}, \bibinfo{author}{C.~S. Calude}, and
  \bibinfo{author}{K.~Svozil}, \bibinfo{journal}{Journal of Mathematical
  Physics} \bibinfo{volume}{\textbf{56}}(10), \bibinfo{pages}{102201(1},
  \bibinfo{eid}{102201} (\bibinfo{date}{2015}), \eprint{arXiv:1503.01985},
  \url{https://doi.org/10.1063/1.4931658}.
\bibitem{svozil-2020-c}
\bibinfo{author}{K.~Svozil}, \bibinfo{journal}{Quantum Reports}
  \bibinfo{volume}{\textbf{2}}(2), \bibinfo{pages}{278} (\bibinfo{date}{2020}),
  \eprint{arXiv:1808.00813}, \url{https://doi.org/10.3390/quantum2020018}.
\bibitem{Gleason}
\bibinfo{author}{A.~M. Gleason}, \bibinfo{journal}{Journal of Mathematics and
  Mechanics (now Indiana University Mathematics Journal)}
  \bibinfo{volume}{\textbf{6}}(4), \bibinfo{pages}{885} (\bibinfo{date}{1957}),
  \url{https://doi.org/10.1512/iumj.1957.6.56050}.
\bibitem{book:1170675}
\bibinfo{author}{E.~Schr{\"{o}}dinger}, \bibinfo{title}{\emph{My View of the
  World}} (\bibinfo{publisher}{Cambridge University Press}, Cambridge, UK,
  \bibinfo{year}{1951}), ISBN \bibinfo{isbn}{9780521062244,9781107049710},
  \url{https://doi.org/10.1017/CBO9781107049710}.
\bibitem{sep-sufficient-reason}
\bibinfo{author}{Y.~Y. Melamed} and \bibinfo{author}{M.~Lin}, in
  \bibinfo{editors}{E.~N. Zalta}, ed., \emph{The {S}tanford Encyclopedia of
  Philosophy} (\bibinfo{publisher}{Metaphysics Research Lab, Stanford
  University}, \bibinfo{year}{2020}), spring 2020 ed.,
  \url{https://plato.stanford.edu/archives/spr2018/entries/sufficient-reason/}.
\bibitem{DonSvo01}
\bibinfo{author}{N.~Donath} and \bibinfo{author}{K.~Svozil},
  \bibinfo{journal}{Physical Review A} \bibinfo{volume}{\textbf{65}},
  \bibinfo{pages}{044302} (\bibinfo{date}{2002}),
  \eprint{arXiv:quant-ph/0105046},
  \url{https://doi.org/10.1103/PhysRevA.65.044302}.
\bibitem{svozil-2002-statepart-prl}
\bibinfo{author}{K.~Svozil}, \bibinfo{journal}{Physical Review A}
  \bibinfo{volume}{\textbf{66}}, \bibinfo{pages}{044306}
  (\bibinfo{date}{2002}), \eprint{arXiv:quant-ph/0205031},
  \url{https://doi.org/10.1103/PhysRevA.66.044306}.
\bibitem{svozil-2003-garda}
\bibinfo{author}{K.~Svozil}, \bibinfo{journal}{Journal of Modern Optics}
  \bibinfo{volume}{\textbf{51}}, \bibinfo{pages}{811} (\bibinfo{date}{2004}),
  \eprint{arXiv:quant-ph/0308110},
  \url{https://doi.org/10.1080/09500340410001664179}.
\bibitem{2007-tkadlec-svozil-springer}
\bibinfo{author}{K.~Svozil} and \bibinfo{author}{J.~Tkadlec},
  \bibinfo{journal}{Natural Computing} \bibinfo{volume}{\textbf{in print}},
  \bibinfo{pages}{in print} (\bibinfo{date}{2009}),
  \url{https://doi.org/10.1007/s11047-009-9112-5}.
\bibitem{svozil-2019-s}
\bibinfo{author}{K.~Svozil}, in \bibinfo{editors}{L.~Grandinetti, S.~L.
  Mirtaheri, and R.~Shahbazian}, eds., \emph{High-Performance Computing and Big
  Data Analysis} (\bibinfo{publisher}{Springer International Publishing}, Cham,
  \bibinfo{year}{2019}), \bibinfo{volume}{vol. 891 of \emph{Communications in
  Computer and Information Science}}, \bibinfo{pages}{pp. 504--512}, ISBN
  \bibinfo{isbn}{978-3-030-33494-9, 978-3-030-33495-6},
  \eprint{arXiv:1904.08307},
  \url{https://doi.org/10.1007/978-3-030-33495-6\_39}.
\bibitem{svozil-2016-ggs}
\bibinfo{author}{H.~Havlicek} and \bibinfo{author}{K.~Svozil},
  \bibinfo{journal}{Entropy} \bibinfo{volume}{\textbf{20}}(4),
  \bibinfo{pages}{284(5)} (\bibinfo{date}{2018}), \eprint{arXiv:1606.03873},
  \url{https://doi.org/10.3390/e20040284}.
\bibitem{svozil-2016-vector}
\bibinfo{author}{K.~Svozil}, \bibinfo{journal}{Entropy}
  \bibinfo{volume}{\textbf{18}}(5), \bibinfo{pages}{156(1}
  (\bibinfo{date}{2016}), \eprint{arXiv:1601.07106},
  \url{https://doi.org/10.3390/e18050156}.
\bibitem{Farhi-98}
\bibinfo{author}{E.~Farhi}, \bibinfo{author}{J.~Goldstone},
  \bibinfo{author}{S.~Gutmann}, and \bibinfo{author}{M.~Sipser},
  \bibinfo{journal}{Physical Review Letters} \bibinfo{volume}{\textbf{81}},
  \bibinfo{pages}{5442} (\bibinfo{date}{1998}),
  \eprint{arXiv:quant-ph/9802045},
  \url{https://doi.org/10.1103/PhysRevLett.81.5442}.
\bibitem{zeil-99}
\bibinfo{author}{A.~Zeilinger}, \bibinfo{journal}{Foundations of Physics}
  \bibinfo{volume}{\textbf{29}}(4), \bibinfo{pages}{631}
  (\bibinfo{date}{1999}), \url{https://doi.org/10.1023/A:1018820410908}.
\bibitem{Schwinger.60}
\bibinfo{author}{J.~Schwinger}, \bibinfo{journal}{Proceedings of the National
  Academy of Sciences (PNAS)} \bibinfo{volume}{\textbf{46}},
  \bibinfo{pages}{570} (\bibinfo{date}{1960}),
  \url{https://doi.org/10.1073/pnas.46.4.570}.
\bibitem{havlicek-svozil-2020-dec}
 (\bibinfo{date}{2020}), \eprint{arXiv:2010.09506},
  \url{http://arxiv.org/abs/2010.09506}.
\bibitem{v-neumann-55}
\bibinfo{author}{J.~{von Neumann}}, \bibinfo{title}{\emph{Mathematical
  Foundations of Quantum Mechanics}} (\bibinfo{publisher}{Princeton University
  Press}, Princeton, NJ, \bibinfo{year}{1955}), ISBN
  \bibinfo{isbn}{9780691028934}, {G}erman original in~\cite{v-neumann-49},
  \url{http://press.princeton.edu/titles/2113.html}.
\bibitem{london-Bauer-1939}
\bibinfo{author}{F.~London} and \bibinfo{author}{E.~Bauer},
  \bibinfo{title}{\emph{La theorie de l'observation en m\'ecanique quantique;
  {N}o.~775 of Actualit\'es scientifiques et industrielles: Expos\'es de
  physique g\'en\'erale, publi\'es sous la direction de {P}aul {L}angevin}}
  (\bibinfo{publisher}{Hermann}, Paris, \bibinfo{year}{1939}), english
  translation in~\cite{london-Bauer-1983}.

\end{thebibliography}


\title{Vector computation}

\author{Karl Svozil}

\address{Institute for Theoretical Physics, TU Wien,\\
Wiedner Hauptstrasse 8-10/136, 1040 Vienna,  Austria\\
E-mail: svozil@tuwien.ac.at\\
http://tph.tuwien.ac.at/\string~svozil}

\begin{abstract}
Quantum physical resources are directional quantities that can be formalized by unit vectors or the associated orthogonal projection operators. When compared to classical computational states which are elements of (power) sets vector computations offer (dis)advantages.
\end{abstract}

\keywords{Value indefiniteness, Quantum mechanics, Gleason theorem, Kochen-Specker theorem, Born rule, vector computation}

\bodymatter


\section{Epistemology versus ontology in the quantum computation context}

In order to claim practical relevance,
any notion and quantitative means of ``computation'' has to be ultimately grounded in
physics, because \emph{``information is physical''}~\cite{landauer},
and so is the manipulation of information.
The Church-Turing thesis --
in Turing's own words \emph{``a man
provided with paper, pencil, and rubber, and subject to strict discipline, is in
effect a universal machine.''}~\cite{copeland04turing}
--
is a conjecture attempting to achieve just this goal: connecting physics to an appropriate formalism.

Yet such conceptualizations bear, in their very success, a dangerous tendency
to forget about the analogy and instead go for the formalism.
Thereby nature is confused with formalism -- indeed, our own partly formalized narratives about nature
-- just as theater~\cite{Arthaud-en} is taken for life,
or propaganda~\cite{bernais-propaganda} for fact.
This results in beliefs in theoretical and hypothetical conceptual entities,
\emph{``the existence of various religious stigmas and social pressures, that taken
together, amount[[ed]] to an evangelical crusade''}~\cite{clauser-talkvie}.

Thereby, issues related to the formal representability of physical entities and processes
are far from settled, and \emph{``bordering on the mysterious''}~\cite{wigner}.
Yet, unlike Wigner, I believe that there may be at least two handy but mutually contradicting
reasons for the effectiveness of mathematics in the natural sciences: one postulates
that laws ``emerge'' from disorder~\cite{Exner-1908,Yanofsky-chaos,svozil-2018-was}.
Another, apparently converse, reason for lawfulness is that we are inhabiting a virtual reality
simulated by a computational process~\cite{zuse-70,toffoli:79,simula,permutationcity,Bostrom-sim}.
In this hypothesis whatever
``laws of nature'' science might discover should be perceived as epistemic,
intrinsic reflections of, or correspondences to, this ontological, extrinsic (to us) computation.

The above rant served the purpose to suggest, and prepare the Reader for, a cautious disengagement between
epistemology and ontology -- what
might be claimed and believed to be known on the one hand,
and what may be ``lurking behind the detector clicks'' on the other hand.
Even with this proviso, one has to keep in mind that there is no ``Archimedean point''
or ``ontological anchor'' upon which an ``objective reality''
(whatever that is) can be based.

In particular, whenever claims are issued about quantum resources, assets, or features
-- such as quantum parallelism by coherent superposition of classically distinct states,
or entanglement as the relational encoding of multi-partite states --
which might go beyond operationally established classical means and could give rise to quantum advantages,
caution is advisable.
Because in such cases the metaphor might supervene or ``outperform'' the simulated,
yielding improper expectations and overstated, almost evangelical, claims~\cite{svozil-2016-quantum-hokus-pokus}.


\section{Types of quantum oracles for randomness: pure states in a superposition versus mixed states}

\label{2020-vectorc-irrev}

There is a sombre fact of contemporary quantum physics:
as of today the ``measurement problem'', as contemplated by von Neumann~\cite{v-neumann-49},
Schr\"odinger~\cite{schrodinger,schroedinger-interpretation},
and repeated by Everett~\cite{everett,everett-1956} and Wigner~\cite{wigner:mb},
despite numerous attempts to resolve it~\cite{london-Bauer-1983},
remains disputed and unsolved.
One of the issues is the simple mathematical impossibility to achieve irreversibility
-- the notorious ``collapse of the wave function'', or state reduction at measurement point~\cite{mermin-07} --
from a completely reversible temporal evolution.
Indeed this should be indisputable from ``group theory~101'': the concatenation of unitary operators never yields outside of the realm of unitary and thus reversible transformations.
For finite groups, Cayley's theorem affirms %$$$$$$$$$$$$
that one essentially is dealing with permutations,
with one-to-one transformations, with re-expressions, re-samplings of the same ``message''.

Stated differently a nesting argument~\cite{v-neumann-49,everett,everett-1956,wigner:mb} essentially ``enlarges'' the domain of reversibility to
include whatever resources or regions of (Hilbert or configuration) space are necessary to re-establish reversibility,
thereby disputing any irreversibility postulated
by quantum mechanics~\cite{PhysRevD.22.879,PhysRevA.25.2208,greenberger2,Nature351,Zajonc-91,PhysRevA.45.7729,PhysRevLett.73.1223,PhysRevLett.75.3783,hkwz}.
For all practical purposes~\cite{bell-a} quantum systems remain ``epistemically'' irreversible~\cite{engrt-sg-I,engrt-sg-II},
but so are classically reversible statistical systems, for which the entropy increase dissolves into thin air
as one looks ``closer'' at individual constituents~\cite{Myrvold2011237}.

Having just avoided the quantum Scylla of ``irreversibility through reversibility''
brings us closer to the quantum Charybdis of ``quantum jellification''
by the prevailing quantum superposition of classically distinct states of matter and mind,
described so vividly in the late Schr\"odinger's Dublin seminars~\cite{schroedinger-interpretation}, repeating the cat paradox~\cite{schrodinger}
in terms of jellyfish.
Without measurement -- how can we and everything around us ``remain stable'' and \emph{not} dissolve into the chasm of coherent superposition,
and how come that, despite this, we experience a fairly unique cognition and presence?

Dirac's ``why bother?'' objection~\cite{bell-a} to all of this might be to not bother at the moment and
leave the task of solving these issues to future generations. Feynman went a step further
and demanded to cease thinking about them, thereby taking the formalism for granted (like a gospel) as given,
and thus effectively shut-up and calculate~\cite{feynman-law,mermin-1989-shutup,mermin-2004-shutup}.

However, such a repression strategy has consequences:
first of all, as mentioned earlier the deterministic, unitary evolution of quantum states (through nesting) contradicts assumptions about irreversible measurements.
Thereby the ontology of the alleged irreducible randomness of single events through measurements of a coherent superposition of classically
distinct or even mutually exclusive states remains open.
Under such circumstances no justification for their stochastic character, no ``quantum certification''~\cite{zeil-05_nature_ofQuantum}
of, in theologic terminology, \emph{creatio continua},
can be given, because consistency issues are unresolved and currently abolished, thereby either relegating a resolution of the argument to the future
or not addressing the ontology at all.
Quantum uncertainty and random outcomes via irreversible measurements of coherent superposition therefore remains conjectural.

Another related ontological issue regards the existence of mixed states.
First of all, the same issues as for measurements are pertinent: how does one obtain a mixed state from pure ones by
one-to-one unitary state evolution?
Claiming to be able to obtain mixed states from pure ones amounts to pretending to get along with outright mathematical incorrectness.

One formal way of generating mixed states from pure multi-partite states
is by taking the partial trace with respect to the Hilbert space of one particle,
a ``beam dump'' of sorts\cite{engrt-sg-I,engrt-sg-II}.
As the trace is essentially a many-to-one operation, irreversibility ensues.
This can be easily corroborated by the non-uniqueness of purification which can be envisioned as the ``quasi-inverse''
(because, strictly speaking, due to non-uniqueness there is no inverse) of the partial trace~\cite[Section~8.3.1]{nielsen-book10}.

So what is the ontologic status of mixed quantum states? This is again unknown and may also, %$$$$$$$$$$$$$$$$$$$
by the  ``why bother?'' objection~\cite{bell-a},
be relegated until some later times.
Again the question arises: how can we trust quantum random sequences originating from mixed state measurements?

Even more convoluted is the question if there is any criterion or difference between sequences generated
by measurement of pure states which are in a coherent superposition on the one hand,
and by measurement of mixed quantum states on the other hand.
As we cannot even formally grasp consistently the notions of irreversible measurement and production of mixed states
we ought to accept our total loss to conceptualize the differences or similarities between them.
We are left with the hope that, as both issues appear to have the same group-theoretic roots,
any solution of one will, by reduction, entail a solution of the other.


Another ``fashionable'' attempt to ascertain and thereby certify quantum randomness involving ``quantum contextuality''~\cite{svozil-2009-howto,svozil-2017-b}
would be to assume the co-existence of complementary observables
and, relative to the respective implicit assumptions (most notably context independence), prove (by contradiction or statistical demonstration) the impossibility
for any value definiteness of complementary, incompatible observables prior to measurements~\cite{kochen1,froissart-81,pitowsky:218,2015-AnalyticKS,svozil-2017-b}.
Suffice it to say that, as these hypothetical arguments go they are contingent on the respective
counterfactual configurations imagined, and thus appear subjective and inconclusive~\cite{svozil-2020-c}.


\section{Questionable parallelism by views on a vector}

From now on only pure states, represented by (unit) vectors spanning a one-dimensional subspace of a vector space, will be considered.
Schr\"odinger's aforementioned question regarding quantum jellification~\cite{schroedinger-interpretation},
a variant of his earlier ``cat paradox''~\cite{schrodinger}, might be brought to  practical use for quantum parallelization:
for the sake of irritation suppose, as is often alleged in quantum computations, that the many mutually exclusive (classical) states in a coherent superposition
\emph{``be not alternatives but all really happen simultaneously $\ldots$
if the laws of nature took this form for, let
me say, a quarter of an hour, we should find our surroundings rapidly
turning into a quagmire, or sort of a featureless jelly or plasma, all contours
becoming blurred, we ourselves probably becoming jelly fish.''}

One of the simplest answers that can be given to these concerns is that this is just an epistemic question arising from a ``wrong'' viewpoint or perspective.
Because if one chooses an orthonormal basis of Hilbert space of which the state vector is an element then
the coherent superposition reduces to a single, unique term -- namely that unit vector.
Unitary quantum evolution amounts to ``rotating'' this vector in Hilbert space.
There is no ``jellification'' of this state with respect to the ``proper'' bases (for dimensions higher than 2 there is a continuum of bases)
of which the state is one part.
In this view, there is no ``simultaneous existence'' of two or more classically distinct states.
All other probabilistic views are mere (continuity of) multiplicities, projections of sorts.
For a similar, more formalized, vision see Gleason's remarks on taking
\emph{``the square of the norm of the projection''} in the second paragraph of Ref.~\cite{Gleason}.

Indeed, one can find some hints on this solution in Schr\"odinger's own writings on {\it the Vedantic vision}~\cite[chapter~V]{book:1170675}:
\emph{``the plurality that we perceive is only \emph{an appearance; it is not real.}
Vedantic philosophy, in which this is a
fundamental dogma has sought to clarify it by a number of
analogies, one of the most attractive being the many-faceted crystal which, while showing hundreds of little
pictures of what is in reality a single existent object, does
not really multiply that object.''
}

This might be considered bad news for quantum parallelism.
Because if any such parallelism is based upon epistemology
-- about appearance without any substantial reality --
all that remains as a resource is our ability to measure properties of the vector
from a ``different angle'' than the one this vector has been defined.

This is particularly pertinent for the ``extraction'' of information in a coherent superposition
by classical irreversible measurements
(cf. my earlier comments in Section~\ref{2020-vectorc-irrev})
``reducing'' a coherent superposition of a exponential variety of hypothetically conceivable classical distinct states
to a \emph{single} such state.
Because what good is it to contemplate about such a counterfactual variety if we have no direct access to it?

It gets even more problematic as we have postulated that the ``extraction'',
the outcome corresponding to this single state, occurs without sufficient reason~\cite{sep-sufficient-reason} and
thereby eventualizes in an irreducibly stochastic~\cite{zeil-05_nature_ofQuantum} manner,
a situation denoted in theology by {\it creatio continua}:
one can get only a single click or outcome per experiment, corresponding to an ``exponential reduction'' of computational states with respect to the state vector representation
prior to extraction.

\section{Computation by projective measurements of partitioned state space}

Nevertheless, even in this reduced scheme of parallelism,
it might be possible to formulate relational queries corresponding to useful information
by appropriate partitioning of the state space~\cite{DonSvo01,svozil-2002-statepart-prl,svozil-2003-garda}.
We may formulate the fundamental problem of intrinsically operational vector encoding of a computation aka quantum computation:
\emph{under what circumstances is it possible to derive ``useful'' information about (the components) of a vector?}


It is not too unreasonable to suspect that an answer to this question can be given in terms
of \emph{relational properties} of the vector encoding~\cite{2007-tkadlec-svozil-springer,svozil-2019-s}.
Deutsch's algorithm~\cite[Section~1.4.3]{nielsen-book10} as well as the quantum Fourier transform
based on period finding~\cite[Section~5.4.1]{nielsen-book10} may be examples for such relational encodings
realizable by state mismatches (``prepare one state, measure another'').
More generally, the partitioning of finite groups by cosets, in particular, the hidden subgroup problem~\cite[Section~5.4.3]{nielsen-book10}
maybe a way to systematically exploit views on vectors,
but so far there exists only anecdotal evidence~\cite[Figure 5.5., p.~241]{nielsen-book10} for that.


There is a way to extract information from a quantum state by constructing proper (with respect to the computational task or query) subspaces
and the orthogonal projection operators onto such subspaces.
How ought this computational method be understood?
It will be argued that it could be perceived in terms of (equi)partitioning the quantum state space,
and, as mentioned earlier,
by using the respective projection operators as filters~\cite{DonSvo01,svozil-2002-statepart-prl,svozil-2003-garda}.

Suppose it is possible to encode (the solution to) a problem into a Hilbert space spanned by a
collection of pure state vectors encoding functional instances and functional properties into one- and higher-dimensional subspaces thereof.
For instance, a binary function of several bits can be encoded into quantum states
(i) by state vectors whose first components with respect to an orthonormal basis (normalization aside) are either 0 or 1,
depending on whether or not the function on the inputs evaluates to 0 or 1, and
(ii)
later ``auxiliary'' components are added to ensure mutual orthogonality of the state vectors.
That is, in order to obtain an orthonormal basis one could, for instance, employ dimensional lifting, and thereby enlarge the Hilbert space.
This results in an orthonormal (after normalization) basis of the aforementioned Hilbert space.
Therefore the elements of the orthonormal basis represent the individual instances of the function or problem.

Now if one forms the orthogonal projection operator
as the sum of the dyadic products of the respective vectors of the orthonormal basis of the subspace encoding a particular problem or a query,
then this projection operator is capable of solving the property or problem it was encoded to solve in a single run.
All that needs to be done is apply this projection ``filter'' to a state encoding some arbitrary problem instance.

Let me demostrate this by an example which is a generalized Deutsch algorithm.
Consider arbitrary binary functions of $n$ classical bits.
Suppose an unknown arbitrary such function is given, and suppose that the question is not which function exactly it is,
but about a relational property which for instance refers to ``common'' or ``different'' properties of functions of this class; say parity.

How could one find such a particular property without having to identify the respective function completely?
It is not too difficult to argue that there are $2^n$ possible arguments and $2^{2^n}$ such binary functions of $n$ bits.
For the sake of a reasonable ``small'' demonstration, take $n=2$
($n=1$ amounts to Deutsch's problem; cf. Refs.~\cite[Section~2.2]{mermin-07} and~\cite[Section~1.4.3]{nielsen-book10}).
Table~\ref{2020-vectorcomputation-table1} enumerates all binary functions of two classical bits.
(At this point we are not dealing with questions of enlarging the Hilbert space to obtain overall reversibility in case the functions are not reversible.)



In the next step, a system of vectors $\vert {\bf e}_i \rangle$
is obtained by identifying the valuations of the functions on the respective bit values with entries in coordinate tuples,
as enumerated in the second to last column of Table~\ref{2020-vectorcomputation-table1}.
Based on these vectors an orthonormal basis of a (subspace) of a high-dimensional Hilbert space can
be effectively generated by dimensional lifting~\cite{svozil-2016-ggs}, so that
$\vert {\bf e}_i    \rangle \mapsto \vert {\bf b}_i    \rangle$ with $\langle  {\bf b}_i    \vert {\bf b}_j    \rangle =\delta_{ij}$ and $i,j=1,\ldots ,16$
as enumerated (without normalization) in the last column of Table~\ref{2020-vectorcomputation-table1}.
Thereby the zero vector $\vert {\bf e}_1    \rangle$ can, for instance, be ad hoc mapped into a
subspace of this larger dimensional Hilbert space which is orthogonal to all other subspaces
(this may be achieved by adding another dimension).
Note that
\begin{itemize}
\item[(i)] in general dimensional lifting is neither unique --
there exit other, rather inefficient methods~\cite{svozil-2016-vector} (with respect to the number of auxiliary extra dimensions)
to orthogonalize the vectors corresponding to the functions $f_i$.
\item[(ii)] Dimensional lifting does not correspond to a unitary transformation as it intentionally changes the inner products (to become zero) in transit to higher dimensions.
Therefore, if one attempts to encode this kind of problems into orthogonal bases of subspaces of higher dimensional Hilbert spaces
one needs to take care of orthogonality from the very beginning.
That is, there has to be a physically feasible way to map the functions $f_i$ into $\vert {\bf b}_i \rangle$.
\item[(iii)]
Accordingly, any way to map the 16 functions $f_i$ into any kind of system of orthogonal vectors suffices for this method as long as it is physically feasible.
\end{itemize}

In the final step a filter is designed which models the binary question by projecting the answers onto the appropriate subspace of the (sub)space spanned by the orthonormal
basis $\big\{ \vert {\bf b}_1 \rangle , \ldots ,  \vert  {\bf b}_{16}   \rangle   \big\}$
such that the question can be answered in a single query.


\begin{sidewaystable}
\vspace*{+250pt}
\tbl{The 16 binary functions of two classical bits. Dots as vector components represent ``very large'' numbers.\label{2020-vectorcomputation-table1}}
{\normalsize
%\resizebox{\textwidth}{!}{
\begin{tabular}{ccccccc}
\toprule
$f_\#$ &00&01&10&11 & corresponding vector & vector after dimensional lifting (20 dimensions) \\
\midrule
$f_1   $&0  &  0 &   0 &   0  & $\vert {\bf e}_1    \rangle=\begin{pmatrix} 0  ,  0 ,   0 ,   0  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_1    \rangle=\begin{pmatrix}0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                           $}\\
$f_2   $&0  &  0 &   0 &   1  & $\vert {\bf e}_2    \rangle=\begin{pmatrix} 0  ,  0 ,   0 ,   1  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_2    \rangle=\begin{pmatrix}0 , 0 , 0 , 1 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                           $}\\
$f_3   $&0  &  0 &   1 &   0  & $\vert {\bf e}_3    \rangle=\begin{pmatrix} 0  ,  0 ,   1 ,   0  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_3    \rangle=\begin{pmatrix}0 , 0 , 1 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                           $}\\
$f_4   $&0  &  0 &   1 &   1  & $\vert {\bf e}_4    \rangle=\begin{pmatrix} 0  ,  0 ,   1 ,   1  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_4    \rangle=\begin{pmatrix}0 , 0 , 1 , 1 , 0 , -1 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                         $}\\
$f_5   $&0  &  1 &   0 &   0  & $\vert {\bf e}_5    \rangle=\begin{pmatrix} 0  ,  1 ,   0 ,   0  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_5    \rangle=\begin{pmatrix}0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                           $}\\
$f_6   $&0  &  1 &   0 &   1  & $\vert {\bf e}_6    \rangle=\begin{pmatrix} 0  ,  1 ,   0 ,   1  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_6    \rangle=\begin{pmatrix}0 , 1 , 0 , 1 , 0 , -1 , 0 , -2 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                        $}\\
$f_7   $&0  &  1 &   1 &   0  & $\vert {\bf e}_7    \rangle=\begin{pmatrix} 0  ,  1 ,   1 ,   0  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_7    \rangle=\begin{pmatrix}0 , 1 , 1 , 0 , 0 , 0 , -1 , -2 , -1 , -6 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                       $}\\
$f_8   $&0  &  1 &   1 &   1  & $\vert {\bf e}_8    \rangle=\begin{pmatrix} 0  ,  1 ,   1 ,   1  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_8    \rangle=\begin{pmatrix}0 , 1 , 1 , 1 , 0 , -1 , -1 , -4 , -1 , -12 , -84 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                   $}\\
$f_9   $&1  &  0 &   0 &   0  & $\vert {\bf e}_9    \rangle=\begin{pmatrix} 1  ,  0 ,   0 ,   0  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_9    \rangle=\begin{pmatrix}1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                           $}\\
$f_{10}$&1  &  0 &   0 &   1  & $\vert {\bf e}_{10} \rangle=\begin{pmatrix} 1  ,  0 ,   0 ,   1  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_{10} \rangle=\begin{pmatrix}1 , 0 , 0 , 1 , 0 , -1 , 0 , -2 , 0 , -6 , -40 , -3442 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                                 $}\\
$f_{11}$&1  &  0 &   1 &   0  & $\vert {\bf e}_{11} \rangle=\begin{pmatrix} 1  ,  0 ,   1 ,   0  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_{11} \rangle=\begin{pmatrix}1 , 0 , 1 , 0 , 0 , 0 , -1 , -2 , 0 , -4 , -30 , -2578 , -1 , -8874706 , 1 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                          $}\\
$f_{12}$&1  &  0 &   1 &   1  & $\vert {\bf e}_{12} \rangle=\begin{pmatrix} 1  ,  0 ,   1 ,   1  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_{12} \rangle=\begin{pmatrix}1 , 0 , 1 , 1 , 0 , -1 , -1 , -4 , 0 , -10 , -70 , -6020 , -1 , -20723712 , \cdot , 1 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal                   $}\\
$f_{13}$&1  &  1 &   0 &   0  & $\vert {\bf e}_{13} \rangle=\begin{pmatrix} 1  ,  1 ,   0 ,   0  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_{13} \rangle=\begin{pmatrix}1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , -1 , -2 , -14 , -1202 , -1 , -4137858 , \cdot , \cdot , 1 , 0 , 0 , 0 \end{pmatrix}^\intercal                   $}\\
$f_{14}$&1  &  1 &   0 &   1  & $\vert {\bf e}_{14} \rangle=\begin{pmatrix} 1  ,  1 ,   0 ,   1  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_{14} \rangle=\begin{pmatrix}1 , 1 , 0 , 1 , 0 , -1 , 0 , -2 , -1 , -8 , -54 , -4644 , -1 , -15986864 , \cdot , \cdot , \cdot , 1 , 0 , 0 \end{pmatrix}^\intercal            $}\\
$f_{15}$&1  &  1 &   1 &   0  & $\vert {\bf e}_{15} \rangle=\begin{pmatrix} 1  ,  1 ,   1 ,   0  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_{15} \rangle=\begin{pmatrix}1 , 1 , 1 , 0 , 0 , 0 , -1 , -2 , -1 , -6 , -44 , -3780 , -1 , -13012562 , \cdot ,\cdot , \cdot , \cdot , 1 , 0 \end{pmatrix}^\intercal         $}\\
$f_{16}$&1  &  1 &   1 &   1  & $\vert {\bf e}_{16} \rangle=\begin{pmatrix} 1  ,  1 ,   1 ,   1  \end{pmatrix}^\intercal$&{\scriptsize$\vert {\bf b}_{16} \rangle=\begin{pmatrix}1 , 1 , 1 , 1 , 0 , -1 , -1 , -4 , -1 , -12 , -84 , -7222 , -1 , -24861568 , \cdot , \cdot , \cdot , \cdot , \cdot , 1 \end{pmatrix}^\intercal  $}\\
\bottomrule
\end{tabular} }
% }
%\vspace*{+13pt}
\end{sidewaystable}



Suppose the question is to find the parity of a function $f_i(x,y) \in \{0,1\})$ with $x,y \in \{0,1\}$, $i\in \{1,\ldots ,16\}$.
All we need to do is to partition the functional space
$\big\{   f_1  , \ldots ,   f_{16}   \big\}$
into functions with an even or an odd number of outputs ``1''. More explicitly, for the functions enumerated in Table~\ref{2020-vectorcomputation-table1} and for parity,
the partition is
\begin{equation}
 \Big\{
\big\{
 f_2, f_3,f_5,f_8,f_9,f_{12},f_{14},f_{15}
\big\}
,
\big\{
 f_1, f_4,f_6,f_7,f_{10},f_{11},f_{13},f_{16}
\big\}
\Big\}
,
\end{equation}
corresponding to the orthogonal projection operators
\begin{equation}
\begin{split}
\textsf{\textbf{E}}_1 =
 \vert \mathbf{b}_2 \rangle \langle \mathbf{b}_2 \vert
+\vert \mathbf{b}_3 \rangle \langle \mathbf{b}_3 \vert
+\vert \mathbf{b}_5 \rangle \langle \mathbf{b}_5 \vert
+\vert \mathbf{b}_8 \rangle \langle \mathbf{b}_8 \vert \qquad \qquad \\
+\vert \mathbf{b}_9 \rangle \langle \mathbf{b}_9 \vert
+\vert \mathbf{b}_{12} \rangle \langle \mathbf{b}_{12} \vert
+\vert \mathbf{b}_{14} \rangle \langle \mathbf{b}_{14} \vert
+\vert \mathbf{b}_{15} \rangle \langle \mathbf{b}_{15} \vert
,\text{ and }\\
\textsf{\textbf{E}}_0 =  \textsf{\textbf{1}} - \textsf{\textbf{E}}_1 =
 \vert \mathbf{b}_1 \rangle \langle \mathbf{b}_1 \vert
+\vert \mathbf{b}_4 \rangle \langle \mathbf{b}_4 \vert
+\vert \mathbf{b}_6 \rangle \langle \mathbf{b}_6 \vert
+\vert \mathbf{b}_7 \rangle \langle \mathbf{b}_7 \vert \qquad \qquad \\
+\vert \mathbf{b}_{10} \rangle \langle \mathbf{b}_{10} \vert
+\vert \mathbf{b}_{11} \rangle \langle \mathbf{b}_{11} \vert
+\vert \mathbf{b}_{13} \rangle \langle \mathbf{b}_{13} \vert
+\vert \mathbf{b}_{16} \rangle \langle \mathbf{b}_{16} \vert
.
\end{split}
\end{equation}
The parity of an unknown given binary function of two bits
can be obtained by a single query measuring the
propositional ``parity property''
associtated with the observable $\textsf{\textbf{E}}_1  = \textsf{\textbf{1}} - \textsf{\textbf{E}}_0$.
In principle this method can be generalized to the parity problem of binary functions of $n$ bits,
utilizing a parallelization of the order of $2^{2^n}$ at the cost of expanding Hilbert space to about twice this number of dimensions.
It remains to be seen whether this method violates the assumptions in Ref.~\cite{Farhi-98}.
In any case it should be noted that parity is an example of a much wider problem class associated with relational properties
which can be represented or parametrized by partitioning appropriate subspaces of Hilbert space.





\section{Entanglement as relational parallelism across multi-partite states}

From a purely formal point of view, entangled particles are modeled by the
indecomposability of state vectors in a Hilbert space which is a non-trivial
tensor product of two or more Hilbert spaces.
Indecomposability means that the respective state vector cannot be decomposed into
a single product of factors of the states of the constituent particles.
Instead, an entangled state can be written as the coherent superposition (aka linear combination) of such product states.

This immediately suggests that whenever such indecomposable vectors occur they can be ``rotated'' by unitary transformations
into a single product form; say a vector of the Cartesian standard basis in the respective tensor product of two or more Hilbert spaces.
Any such transformation cannot be expected to be acting ``locally'' in a single constituent space but rather ``globally''
across the single constituent spaces.

Physically this means that we are not dealing with single-particle properties but again with \emph{relational properties};
whereby relational information is encoded across the multiple constituents of such a state~\cite{schrodinger,zeil-99}.
This can be expected: as unitary transformations are defined by
rotations transforming some orthonormal basis into another orthonormal basis~\cite{Schwinger.60}
this amounts to a kind of ``zero-sum game'' between \emph{localized} information aka properties on \emph{individual} constituents
on the one hand, and  \emph{relational} information which for instance refers to ``common'' or ``different'' properties within
\emph{collectives or groups} of constituents on the other hand.


\section{On partial views of vectors}


In the context of state purification the following more general question arises:
What kind of value might a partial knowledge of or about a vector have?
After all, embedded observers~\cite{toffoli:79} may obtain only partial knowledge and control
of the degrees of freedom entailed by overseeing a ``small'' subset of a ``much larger'' Hilbert space they inhabit.

Suppose an observer has acquired knowledge about an incomplete list of components (relative to a particular basis) of a pure state vector.
This can be formalized either by a projection of this vector onto a subspace of the Hilbert space
or by ``extraction'' of the coordinates by the respective vectors of the dual basis of the dual space.

One possibility would be to ``complete'' the state vector by various procedures,
such as the aforementioned purification.
Alternatively one might consider the subspace spanned by both the given and the missing basis elements~\cite{havlicek-svozil-2020-dec}.


\section{Summary}


I have presented a revisionist glance at quantum computation as seen from an equally revisionist perspective on quantum theory.
The main  t\'elos, that is, the end, purpose, or goal, of these considerations rests in the emphasis that only with a proper understanding
of the quantum physical resources it is possible to develop a comprehensive theory of quantum information and computation.

For instance, the mere communal or individual canonical believe in ontological quantum randomness -- without mentioning the
implicit assumptions or corroborations which are essentially based upon incapacity; that is the experience that nobody so far has come up with any causes and
necessary and sufficient reasons for quantum outcomes -- suggests that any such claims need be viewed as epistemic, anecdotal and preliminary.

I believe that quantum information and computation are intimately tied to foundational issues.
Therefore, such issues need at least to be kept in mind if one assesses the capacities of quantum systems to store and process information.



%\begin{acknowledgments}
\section*{Acknowledgments}
Parts of this discussion have been inspired by a discussion with Noson Yanofsky about
whether quantum superpositions and entanglements make the universe unboring.


The author acknowledges the support by the Austrian Science Fund (FWF): project I 4579-N and the Czech Science Foundation (GA\v CR): project 20-09869L,
%\end{acknowledgments}


%\bibliographystyle{ws-procs9x6}
% \bibliographystyle{revtex}
% \bibliography{svozil}




\end{document}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


ee={{0  ,  0 ,   0 ,   0 },
{0  ,  0 ,   0 ,   1 },
{0  ,  0 ,   1 ,   0 },
{0  ,  0 ,   1 ,   1 },
{0  ,  1 ,   0 ,   0 },
{0  ,  1 ,   0 ,   1 },
{0  ,  1 ,   1 ,   0 },
{0  ,  1 ,   1 ,   1 },
{1  ,  0 ,   0 ,   0 },
{1  ,  0 ,   0 ,   1 },
{1  ,  0 ,   1 ,   0 },
{1  ,  0 ,   1 ,   1 },
{1  ,  1 ,   0 ,   0 },
{1  ,  1 ,   0 ,   1 },
{1  ,  1 ,   1 ,   0 },
{1  ,  1 ,   1 ,   1 }};

ff = Table[{i, j}, {i, 1, 16}, {j, 1, 20}];

Do[ff[[1, j]] =
  If[j <= 4, ee[[1, j]], If[j == 4 + 1, 1, If[j > 4 + 1, 0]]], {j, 1,
  20}]


Do[ff[[i, j]] =
     If[j <= 4, ee[[i, j]],
        If[j == i + 4, 1,
           If[j > i + 4, 0,
              If[j == 5, -ee[[1]].ee[[i]],
                -ee[[j - 4]].ee[[i]] - Sum[   ff[[j - 4, jjj ]]*ff[[i, jjj]]    ,{jjj,5, j-1}]
                ]
             ]
          ]
        ], {j, 1, 20}, {i, 2, 16}]




MatrixForm[ff]

MatrixForm[Table[ff[[i]].ff[[j]], {i, 1, 16}, {j, 1, 16}]]






\left(
\begin{array}{cccccccccccccccccccc}
 0 & 0 & 0 & 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 0 & 0 & 0 & 1 & 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 0 & 0 & 1 & 0 & 0 & 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 0 & 0 & 1 & 1 & 0 & -1 & -1 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 0 & 1 & 0 & 1 & 0 & -1 & 0 & -2 & -1 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 0 & 1 & 1 & 0 & 0 & 0 & -1 & -2 & -1 & -6 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 0 & 1 & 1 & 1 & 0 & -1 & -1 & -4 & -1 & -12 & -84 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\
 1 & 0 & 0 & 1 & 0 & -1 & 0 & -2 & 0 & -6 & -40 & -3442 & -1 & 1 & 0 & 0 & 0 & 0 & 0 & 0 \\
 1 & 0 & 1 & 0 & 0 & 0 & -1 & -2 & 0 & -4 & -30 & -2578 & -1 & -8874706 & 1 & 0 & 0 & 0 & 0 & 0 \\
 1 & 0 & 1 & 1 & 0 & -1 & -1 & -4 & 0 & -10 & -70 & -6020 & -1 & -20723712 & -183916866750384 & 1 & 0 & 0 & 0 & 0 \\
 1 & 1 & 0 & 0 & 0 & 0 & 0 & 0 & -1 & -2 & -14 & -1202 & -1 & -4137858 & -36722276318934 & -6753846000520252085935696594 & 1 & 0 & 0 & 0 \\
 1 & 1 & 0 & 1 & 0 & -1 & 0 & -2 & -1 & -8 & -54 & -4644 & -1 & -15986864 & -141878729835874 & -26093891449938500567239755536 & -176234124407176743759060097368794907923540016498505162876 & 1 & 0 & 0 \\
 1 & 1 & 1 & 0 & 0 & 0 & -1 & -2 & -1 & -6 & -44 & -3780 & -1 & -13012562 & -115482671802964 & -21239211161964327257629367072 & -143446361360437867702271741780900512324900770322654989532 & -25280143893752238193766302269040410690463268132798981063345774266764666786507011316102671817385018451812886397480 & 1 & 0 \\
 1 & 1 & 1 & 1 & 0 & -1 & -1 & -4 & -1 & -12 & -84 & -7222 & -1 & -24861568 & -220639125319904 & -40579256611382575738933426014 & -274066049968871206521884255518225403236609303753979782044 & -48299790345997566151440190433560198025088071637661284848264632792452124976408487397325842205022148918233708534510 & -1221025649984883675985358536224195253438557154538786018715372562898562470321954355419803731053462819341154947203888099621875559538938162404995285432981398927874079411909612099068865705010313467684991159914369123758761656124830 & 1 \\
\end{array}
\right)

\begin{pmatrix}0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}0 , 0 , 0 , 1 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}0 , 0 , 1 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}0 , 0 , 1 , 1 , 0 , -1 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}0 , 1 , 0 , 1 , 0 , -1 , 0 , -2 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}0 , 1 , 1 , 0 , 0 , 0 , -1 , -2 , -1 , -6 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}0 , 1 , 1 , 1 , 0 , -1 , -1 , -4 , -1 , -12 , -84 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}1 , 0 , 0 , 1 , 0 , -1 , 0 , -2 , 0 , -6 , -40 , -3442 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}1 , 0 , 1 , 0 , 0 , 0 , -1 , -2 , 0 , -4 , -30 , -2578 , -1 , -8874706 , 1 , 0 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}1 , 0 , 1 , 1 , 0 , -1 , -1 , -4 , 0 , -10 , -70 , -6020 , -1 , -20723712 , \cdot , 1 , 0 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , -1 , -2 , -14 , -1202 , -1 , -4137858 , \cdot , \cdot , 1 , 0 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}1 , 1 , 0 , 1 , 0 , -1 , 0 , -2 , -1 , -8 , -54 , -4644 , -1 , -15986864 , \cdot , \cdot , \cdot , 1 , 0 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}1 , 1 , 1 , 0 , 0 , 0 , -1 , -2 , -1 , -6 , -44 , -3780 , -1 , -13012562 , \cdot ,\cdot , \cdot , \cdot , 1 , 0 \end{pmatrix}^\intercal
\begin{pmatrix}1 , 1 , 1 , 1 , 0 , -1 , -1 , -4 , -1 , -12 , -84 , -7222 , -1 , -24861568 , \cdot , \cdot , \cdot , \cdot , \cdot , 1 \end{pmatrix}^\intercal

 0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 0 , 0 , 0 , 1 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 0 , 0 , 1 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 0 , 0 , 1 , 1 , 0 , -1 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 0 , 1 , 0 , 1 , 0 , -1 , 0 , -2 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 0 , 1 , 1 , 0 , 0 , 0 , -1 , -2 , -1 , -6 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 0 , 1 , 1 , 1 , 0 , -1 , -1 , -4 , -1 , -12 , -84 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 0 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , 0
 1 , 0 , 0 , 1 , 0 , -1 , 0 , -2 , 0 , -6 , -40 , -3442 , -1 , 1 , 0 , 0 , 0 , 0 , 0 , 0
 1 , 0 , 1 , 0 , 0 , 0 , -1 , -2 , 0 , -4 , -30 , -2578 , -1 , -8874706 , 1 , 0 , 0 , 0 , 0 , 0
 1 , 0 , 1 , 1 , 0 , -1 , -1 , -4 , 0 , -10 , -70 , -6020 , -1 , -20723712 , -183916866750384 , 1 , 0 , 0 , 0 , 0
 1 , 1 , 0 , 0 , 0 , 0 , 0 , 0 , -1 , -2 , -14 , -1202 , -1 , -4137858 , -36722276318934 , -6753846000520252085935696594 , 1 , 0 , 0 , 0
 1 , 1 , 0 , 1 , 0 , -1 , 0 , -2 , -1 , -8 , -54 , -4644 , -1 , -15986864 , -141878729835874 , -26093891449938500567239755536 , -176234124407176743759060097368794907923540016498505162876 , 1 , 0 , 0
 1 , 1 , 1 , 0 , 0 , 0 , -1 , -2 , -1 , -6 , -44 , -3780 , -1 , -13012562 , -115482671802964 , -21239211161964327257629367072 , -143446361360437867702271741780900512324900770322654989532 , -25280143893752238193766302269040410690463268132798981063345774266764666786507011316102671817385018451812886397480 , 1 , 0
 1 , 1 , 1 , 1 , 0 , -1 , -1 , -4 , -1 , -12 , -84 , -7222 , -1 , -24861568 , -220639125319904 , -40579256611382575738933426014 , -274066049968871206521884255518225403236609303753979782044 , -48299790345997566151440190433560198025088071637661284848264632792452124976408487397325842205022148918233708534510 , -1221025649984883675985358536224195253438557154538786018715372562898562470321954355419803731053462819341154947203888099621875559538938162404995285432981398927874079411909612099068865705010313467684991159914369123758761656124830 , 1


